\documentclass{65240}
\input{preamble}
\input{macro}

\author{Your Name}
\pset{1}
\collab{Firstname Lastname}
\due{September 18, 2024}
%or \collab{none}



\begin{document}

\paragraph{Homework guidelines:}
You may work with other students, as long as (1) they
have not yet solved the problem, (2) you write
down the names of all other students with which you
discussed the problem, and (3) you write up the solution
on your own.  No points will be deducted, no matter how
many people you talk to, as long as you are honest.
If you already knew the answer
to one of the problems (call these ``famous" problems),
then let us know that in your solution writeup -- it
will not affect your score, but will help us in the future.
It's ok to look up famous sums and inequalities that help
you to solve the problem, but don't look up an entire solution.

\begin{enumerate}
\item
\end{enumerate}
   

\end{document}
